Author's response to reviews

Title: The persisting burden of invasive pneumococcal disease in HIV patients: an observational cohort study

Authors:

Reed A.C. Siemieniuk (reed.siemieniuk@gmail.com)
Dan B. Gregson (dan.gregson@cls.ab.ca)
M. John Gill (john.gill@albertahealthservices.ca)

Version: 5 Date: 10 November 2011

Author's response to reviews:

November 9, 2011

Re: The persisting burden of invasive pneumococcal disease in HIV patients: an observational cohort study

Dear Editor:

The authors have no further edits to report.

Thank you for considering our manuscript.

Sincerely,

Reed Siemieniuk

Dan Gregson

John Gill